Author's response to reviews

Title: Response of religious groups to HIV/AIDS as a sexually transmitted infection in Trinidad

Authors:

Gillian L Genrich (glgenrich@yahoo.com)  
Brader A Brathwaite (brada@wow.net)

Version: 5 Date: 25 October 2005

Author's response to reviews:

Dear Editorial Team,

On p. 20, "...into my belief, knowing that God is alive," was deleted, as did not add value to the remaining paragraph.

Thank you very much,

Gillian